Author's response to reviews

Title: Malignant transformation of oral potentially malignant disorders in male: a retrospective cohort study

Authors:

- Pei-Shan Ho (psho@kmu.edu.tw)
- Pai-Li Chen (scalpel58@seed.net.tw)
- Saman Warnakulasuriya (s.warne@kcl.ac.uk)
- Tien-Yu Shieh (tiyush@kmu.edu.tw)
- Yun-Kwan Chen (yukkwa@kmu.edu.tw)
- I-Yueh Huang (iyuhu@kmu.edu.tw)

Version: 9 Date: 17 July 2009

Author's response to reviews: see over
Dear Dr. Matthew Kaiser,

We have received your formatting requirements. The revised manuscript of "Malignant transformation of oral potentially malignant disorders in male: a retrospective cohort study" has been uploaded in the system.

Looking forward to hearing from you soon.

Sincerely Yours,

Pei-Shan Ho, PhD
Assistant Professor
Department of Oral Hygiene
Kaohsiung Medical University